Author's response to reviews

Title: Factors affecting post-embolization fever and liver failure after trans-arterial chemo-embolization in a cohort without background infective hepatitis - a prospective analysis

Authors:

Rohan C Siriwardana (rohansiriwardana@yahoo.com)
Madunil A Niriella (maduniln@yahoo.co.uk)
Anuradha S Dassanayake (anuradhadassa@gmail.com)
Chandika A Liyanage (chandikaliyanage@hotmail.com)
Angappulige Upasena (aupasena@yahoo.com)
Chandra Sirigampala (chandrasirigampala@gmail.com)
Hithanadura J de Silva (hjanakadesilva@gmail.com)

Version: 3
Date: 29 July 2015

Author's response to reviews: see over
Dr. M.A. Niriella

Corresponding author
Final changes made to article:

1. Age at the time of procedure as a risk factor is now included in the univariate analysis in Table 1

2. Age at the time of procedure as a risk factor was not included in the multivariate analysis as it was not significant in the univariate model

3. 95% confidence intervals of OR included in Table 2

4. Ethics statement moved to end of Method section